Author's response to reviews

Title: Prevalence of cognitive impairment in individuals aged over 65 in an urban area: DERIVA study

Authors:

Emiliano Rodriguez-Sanchez (emilianorodriguezsanchez@yahoo.es)
Sara Mora-Simón (saramorasimon@gmail.com)
María C Patino-Alonso (carpatino@usal.es)
Ricardo García-García (rigar@usal.es)
Alfonso Escribano-Hernández (aescribanoh@saludcastillayleon.es)
Luis García-Ortiz (Lgarciao@usal.es)
Mª Victoria Perea-Bartolomé (vperea@usal.es)
Manuel A Gómez-Marcos (magomez@usal.es)

Version: 6 Date: 16 November 2011

Author's response to reviews: see over
Dear Editor:

Following the editorial suggestions, we enclose a new version of our manuscript entitled: “Prevalence of cognitive impairment in individuals aged over 65 in an urban area: DERIVA study”, Re: 9557193555848233, together with our replies to all the issues raised.

Formatting changes to be made:

<table>
<thead>
<tr>
<th>References: We notice reference 36 is currently not cited until after reference 57. Please rearrange your references so that they are cited in consecutive numerical order.</th>
<th>We have insert reference 36 in page 8 line 24 and change to 30 authors in references (“Seshadri et al [36] pointed out, depending on where the “...) and we have removed reference 37( Jessen F,...”) in page 9 line 7 publication of new data [ 2, 24, 34, 35, 37].</th>
</tr>
</thead>
<tbody>
<tr>
<td>Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, &quot;The Title of My Prestigious Publication: A Scientific Study.&quot; should be changed to &quot;The title of my prestigious publication: a scientific study&quot;</td>
<td>We have removed this sentence on title. “The importance of certain personal factors”. We think is not necessary.</td>
</tr>
<tr>
<td>References: the reference list should contain up to 30 authors' names for each citation. The term &quot;et al.&quot; should not be used if there are fewer than 30 authors.</td>
<td>we have changed style Endnote.</td>
</tr>
<tr>
<td>Figures: Please ensure all mention of figures is consistent (it should be Figure or Fig but not a mixture of both)</td>
<td>“Figure” (not Fig.)</td>
</tr>
<tr>
<td></td>
<td>We have removed on title table 3: CIND or dementia. we added at the foot of Table 3: CIND or dementia (CI: cognitive impairment (CIND or dementia))</td>
</tr>
</tbody>
</table>

Please contact us if you have any problems or questions.

Best wishes,

Emiliano Rodriguez.